Author’s response to reviews

Title: Malignancy risk in Australian rheumatoid arthritis patients treated with anti-tumour necrosis factor therapy: an update from the Australian Rheumatology Association Database (ARAD) prospective cohort study

Authors:

Margaret Staples (margaret.staples@monash.edu)

Lyn March (lyn.march@sydney.edu.au)

Catherine Hill (Catherine.Hill@sa.gov.au)

Marissa Lassere (m.lassere@unsw.edu.au)

Rachelle Buchbinder (rachelle.buchbinder@monash.edu)

Version: 3 Date: 20 Nov 2018

Author’s response to reviews:

Dear Ms Wicker

The full names of all ethics committees from whom ethical approval was granted has been added as requested to the Ethics Approval and Consent to participate section of your declarations.

We look forward to hearing from you.

Best wishes

Rachelle Buchbinder